Author's response to reviews

Title: A primary care, multi-disciplinary disease management program for opioid-treated patients with chronic non-cancer pain and a high burden of psychiatric comorbidity

Authors:

Paul R Chelminski (chelminp@med.unc.edu)
Timothy J Ives (tijives@med.unc.edu)
Katherine M Felix (claynef@mindspring.com)
Steven D Prakken (sprakken@earthlink.net)
Thomas M Miller (thomas_miller@med.unc.edu)
J Stephen Perhac (john_perhac@med.unc.edu)
Robert M Malone (rmalone@med.unc.edu)
Mary E Bryant (bbryant@med.unc.edu)
Darren A DeWalt (dewaltd@med.unc.edu)
Michael P Pignone (michael_pignone@med.unc.edu)

Version: 5 Date: 16 December 2004

Author's response to reviews: see over
A primary care, multi-disciplinary disease management program for opioid-treated patients with chronic non-cancer pain and a high burden of psychiatric comorbidity

Thank you for your acceptance in principle of our manuscript. We have complied with all of the formatting revisions requested below and have proofread and copy-edited the manuscript.

Sincerely,

Paul Roman Chelminski, MD, MPH

Revisions

Formatting changes requested.

1. Line numbers - Please remove these. Done
2. Affiliations. Done
3. Qualifications - Please remove the authors' qualifications. Done.
4. Please move the following statement to the Acknowledgements section. Done. Please remove the "abstract title".
5. Abstract word count - Please remove the word count from the abstract page. Done.
6. Word count - Please remove the word count. Done.
7. Introduction - Please rename this section 'Background'. Done.
8. References - References must be cited in the text using consecutive numbers in square brackets, not superscript. References - In the manuscript text, please place the reference before the full stop. Done.
9. Section headings - Please remove all unnecessary capitalisation from all headings and sub-headings in the manuscript. Done.
10. Website addresses - Please cite website addresses in the reference list and remove them from the manuscript text. Cite them using the following format: Website title [http://www.website.com]. Done.
11. Competing interests - Please include a 'Competing interests' section between the Conclusions and Authors' contributions. If there are none to declare, please write 'The authors declare that they have no competing interests'. Done. No competing interests.
12. Authors' contributions - Please include an Authors' contributions section between the Competing interests and Acknowledgements. Done.
13. Tables - These should follow the reference list. Done.
14. References - The reference list should contain all authors' names, regardless of how many there are. The term "et al" should not be used. Done.

15. Appendices - Please move these to before the reference list. Done.